

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Eliasson, Tove

## Working Paper Immigrant entrepreneurship and the origin of bankers

Working Paper, No. 2014:19

**Provided in Cooperation with:** IFAU - Institute for Evaluation of Labour Market and Education Policy, Uppsala

*Suggested Citation:* Eliasson, Tove (2014) : Immigrant entrepreneurship and the origin of bankers, Working Paper, No. 2014:19, Institute for Evaluation of Labour Market and Education Policy (IFAU), Uppsala

This Version is available at: https://hdl.handle.net/10419/106269

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



## WWW.ECONSTOR.EU



# Immigrant entrepreneurship and the origin of bankers

**Tove Eliasson** 

WORKING PAPER 2014:19

The Institute for Evaluation of Labour Market and Education Policy (IFAU) is a research institute under the Swedish Ministry of Employment, situated in Uppsala. IFAU's objective is to promote, support and carry out scientific evaluations. The assignment includes: the effects of labour market and educational policies, studies of the functioning of the labour market and the labour market effects of social insurance policies. IFAU shall also disseminate its results so that they become accessible to different interested parties in Sweden and abroad.

IFAU also provides funding for research projects within its areas of interest. The deadline for applications is October 1 each year. Since the researchers at IFAU are mainly economists, researchers from other disciplines are encouraged to apply for funding.

IFAU is run by a Director-General. The institute has a scientific council, consisting of a chairman, the Director-General and five other members. Among other things, the scientific council proposes a decision for the allocation of research grants. A reference group including representatives for employer organizations and trade unions, as well as the ministries and authorities concerned is also connected to the institute.

Postal address: P.O. Box 513, 751 20 Uppsala Visiting address: Kyrkogårdsgatan 6, Uppsala Phone: +46 18 471 70 70 Fax: +46 18 471 70 71 ifau@ifau.uu.se www.ifau.se

Papers published in the Working Paper Series should, according to the IFAU policy, have been discussed at seminars held at IFAU and at least one other academic forum, and have been read by one external and one internal referee. They need not, however, have undergone the standard scrutiny for publication in a scientific journal. The purpose of the Working Paper Series is to provide a factual basis for public policy and the public policy discussion.

ISSN 1651-1166

## Immigrant entrepreneurship and the origin of bankers

#### To ve Eliasson\*†

August 12, 2014

#### Abstract

Earlier research has shown that immigrant- and minority entrepreneurs have difficulties accessing capital through the formal financial markets. This essay studies what role immigrant employees within the local bank sector have for the probability of immigrants to run their own businesses. I use linked employer-employee data covering the whole Swedish labor market for the years 1987 to 2003 and utilize a nationwide refugee dispersal policy to get exogenous variation in the exposure to co-ethnic bank employees. Results suggest that there is a positive relation between co-ethnic bank employees and the probability of being self-employed. This effect is most pronounced for immigrants who arrived with low education, for males and for those residing in metropolitan regions. The effects are substantial and robust to a wide set of controls for labor market characteristics of the ethnic group at the local level. These results provides evidence of an ethnic component in the formal credit markets.

Key words: Self-employment, immigrant entrepreneurs, capital access, information asymmetry, minority representation

JEL codes: G21, M13, J71

<sup>\*</sup>Institute for Housing and Urban Research, Uppsala University and Department of Economics, Uppsala University

<sup>&</sup>lt;sup>†</sup>I am grateful for valuable discussions with Oskar Nordström Skans, Per-Anders Edin, Lori Beaman, Cynthia Kinnan, Per Strömblad, Kristiina Huttunen, Olof Åslund as well as of the comments and suggestions of participants at the IBF Seminar in Uppsala 2013, SUDSWEC conference in Stockholm 2013, Conference on Discrimination and Labour Market Research in Växjö 2013 and seminar at IFAU, Uppsala 2013. All help has been highly appreciated and all errors are my own. Financial support from FAS/FORTE is gratefully acknowledged.

## Contents

1	Intr	oduction	3
<b>2</b>	Cap	ital access for immigrant entrepreneurs	<b>5</b>
	2.1	Lower rates of bank loans	5
	2.2	Co-ethnic bankers and potential capital access	6
3	Dat	a and sample selection	8
	3.1	Defining the explanatory variable	9
	3.2	Descriptive statistics	10
4	Eco	nometric strategy	12
	4.1	OLS specification	12
	4.2	Municipal group level covariates	13
<b>5</b>	OLS	S results	14
6	Ana	lysis based on initial exogenous variation	17
	6.1	Placement of refugees: the policy	17
	6.2	Modeling the impact of placement	18
	6.3	Instrumental variable analysis	19
	6.4	Results based on initial exogenous variation	20
	6.5	The importance of local group quality	23
7	Exte	ensions	<b>23</b>
	7.1	Own-group employees of a similar sector	23
	7.2	Dynamics of self-employment: entry and exit	27
8	Sens	sitivity	29
9	Con	cluding remarks	31
10	App	pendix	36
	10.1	Variable description	36
		10.1.1 List of municipal group level covariates	39
	10.2	Description; bank, insurance and self-employment	40
	10.3	Further regression results	44

## 1 Introduction

Immigrants from low-income source countries struggle to find employment in many western countries. Self-employment may offer an alternative entry way into the labor market for these individuals (Blume et al., 2009; Andersson Joona and Wadensjö, 2008), but starting a business is not without its own constraints. In particular, small businesses are often liquidity constrained,<sup>1</sup> and the relationship with lenders can therefore be crucial for the start-up and survival of small firms. Banks base their credit decisions on both objective credit scores and subjective assessments of the business idea. This has been suggested to contribute to immigrant- and minority entrepreneurs facing discrimination from lenders (Cavaluzzo and Cavaluzzo, 1998; Blanchflower et al., 2003; Bates, 1989).

The aim of this essay is to contribute to the literature by studying the role of local co-ethnic bank employees for the probability of immigrants to run their own businesses. To this end, I use a linked employer–employee data set covering the whole Swedish labor market for the years 1987 to 2003, from which I can define the country of origin of both bank employees and small business owners. This analysis adds to our understanding of the relationship between banks and entrepreneurs. It also provides novel evidence of the role of minority representation in a sector where there is room for subjective personal assessments. To the extent that credit is allocated according to factors unrelated to the potential prospects of the business idea, society loses out on investment and individual entrepreneurs lose a potential income source. For this reason it is important to understand which factors determine capital access for small business owners.

Following Bates (1989), a number of articles have tried to explain group differences in credit acquisition for small firm start-ups and survival.<sup>2</sup> A more recent strand of this literature focuses on how social ties between lenders and entrepreneurs affect the accessed credit, with a specific focus on shared ethnicity or language. Using survey data, both Aaronson et al. (2004) and Raijman and Tienda (2010) show that same-language suppliers are important for trade credit.

The credit decision performed by the bank is also similar to decisions made in a range of areas in which there is room for discretion. These types of decisions have been extensively studied by scholars within the field of minority representation, focusing on contexts such as educational administration, the police and the judicial

<sup>&</sup>lt;sup>1</sup> Evans and Leighton (1989); Evans and Jovanovich (1989); Lindh and Ohlsson (1996, 1998); Blanchflower and Oswald (1998).

 $<sup>^{2}</sup>$ See for example Blanchflower et al. (2003); Fraser (2009); Cavaluzzo and Cavaluzzo (1998); Cavaluzzo and Wolken (2005).

system. Typically, these studies show that minority representation is positively related to minority outcomes (see Bradbury and Kellough (2011) for an overview). For Sweden it has also been shown that immigrant managers are more prone to hire immigrant employees than native managers are, indicating a relatively favorable behavior towards the non-majority group from minority managers (Åslund et al., forthcoming).

There are, however, very few previous studies on how representation in banks affects capital allocation. Fisman et al. (2012) is the main exception to this. They study the personal loan market in India and make use of a rotation of bank managers between regions, which implies a near random assignment of a customer to either a bank official from the own group or from another group. Their results indicate that in-group bank officials perform better ex ante screening, as in-group borrowers have higher repayment rate as well as larger variance of the loan amount. My aim is to contribute with new knowledge regarding the relationship between entrepreneurs and formal banks in an industrialized western context, focusing on credit access for a group which has been shown to have difficulties in many labor markets.

Banks may employ individuals from a specific country in response to the groups' demand for credit. Therefore OLS estimates imply the existence of an ethnic component in the capital market, but they do not tell us the direction of causality. This, along with individuals potentially sorting into municipalities with good banking and business opportunities, may bias the estimates. In an attempt to remedy this bias, I use a nationwide refugee dispersal policy in place in the second half of the 1980s as a source of exogenous settlement.<sup>3</sup> During this time the refugees who arrived to Sweden were placed in municipalities all over the country, thus providing settlement patterns that are exogenous to the outcome. This means that, for the group of refugees who were placed through this policy, we can estimate the effect of the exogenous exposure to co-ethnic bankers their first year in Sweden, as well use initial exposure as an instrument for exposure in later years.

The OLS results show a positive co-variation between co-ethnic bank employees and the probability of being self-employed. Results from specifications using the exogenous arrival year exposure to co-ethnic bankers are in line with the OLS estimates. These results imply that self-employment is facilitated by the exposure to local own-group bank employees. The estimates are largest for low educated, for males and for those residing in metropolitan areas. Highly educated individuals potentially have more relevant experience and credentials which facilitate capital ac-

 $<sup>^{3}</sup>$ It was in place between 1985 and 1994 but its strictest application was between 1987 and 1991 (see institutional background).

cess and entry into the regular labor market. Thus it is likely that the low educated immigrants benefit more from the community's resources in this regard. To ensure that what we estimate is not the effect of a good labor market position of the group at the local level, a wide range of municipal group level covariates are controlled for in the model. A number of sensitivity checks are performed for the whole group and for the group of low educated, and the results are robust to changes in the definitions of the main variables as well as other variables.

The remainder of this essay is structured as follows. The next section describes immigrant entrepreneurship and capital allocation, while section 3 describes the data and the sample, and presents some descriptive statistics. The econometric model is outlined in section 4 and in section 5 presents the baseline result. Section 6 outlines an analysis based on initial exogenous variation in co-ethnic bank representation, and presents results from this analysis. Section 7 presents two extensions of the baseline model, and section 8 presents a sensitivity analysis. Section 9 concludes.

## 2 Capital access for immigrant entrepreneurs

#### 2.1 Lower rates of bank loans

To start-up and run a small business is often entailed with difficulties, many of which are related to the large information asymmetries which complicate banks' credit decisions. The bank clerks have imperfect knowledge regarding the productivity of the entrepreneur, and they therefore carefully assess the default risk before granting a loan. In this process the banks use information on past defaults or employment history, as well as less objective measures such as the bank clerks' personal assessment of the business idea (Fraser, 2009; Committee on Foreign Born Entrepreneurship, 1999).

Regardless the evaluation method, there are factors which might affect the capital decision negatively for immigrants. When asked about barriers for running business, immigrants from countries outside of Europe state that they find it difficult to access the necessary capital far more often than natives and European immigrants do (Agency for Economic and Regional Growth, 2011, 2007). However, part of these differences can be explained by factors such as lack of own business capital and an inclination to start firms in over-established sectors. Also, the formal requirements might act as a barrier for recently arrived immigrants who do not have credit histories on record in the host country. It is not clear to what degree these factors explain the larger perceived difficulty of loan access (Committee on Foreign Born

Entrepreneurship, 1999).

Even immigrants with strong credentials and relevant experiences from the country of origin might face other difficulties than natives do when applying for a loan. Immigrants may not always have proper information regarding the host country financial system. Furthermore, bank clerks may find it more difficult to evaluate a business idea posed by a person with foreign background, both due to cultural differences and the potential language barrier (Committee on Foreign Born Entrepreneurship, 1999). This uncertainty may lead to statistical discrimination, where immigrants face credit decision based more on the perceived productivity of the group, than of the individual (Phelps, 1972). In addition, trust built through personal contacts with credit suppliers are important for reducing credit uncertainty (Fraser, 2009) and immigrants' short history within the host country system makes them less likely to have a well-established contact with bank employees.

Potentially as a consequence of the poor access to financial markets, immigrants finance their businesses to a higher degree with capital obtained from family and friends than native entrepreneurs do. Abbasian and Yazdanfar (2012) show, for a sample of entrepreneurs in Sweden, that the share of immigrant entrepreneurs who has used bank loans as a means of start-up finance is 29%, compared to 35% for native born entrepreneurs. A governmental report shows that the main financial sources for small business owners differs substantially depending on country of origin (Agency for Economic and Regional Growth, 2007).<sup>4</sup> Immigrants' loan applications also seem to be rejected far more often than natives' applications. Between 8 and 10 percent (depending on year studied) of native entrepreneurs state that their application has been rejected while for the immigrants from refugee countries (i.e. the group I study), between 30-50 % state that they have applied for a loan and been rejected. Interviews with small business owners indicate that some individuals also refrain from applying for loans as they expect to be rejected (Agency for Economic and Regional Growth, 2007).

#### 2.2 Co-ethnic bankers and potential capital access

There are two main reasons why bank employees from the own ethnic group may be positively related to the probability of obtaining capital for small businesses, and hence also for the probability of being self-employed. These individuals may improve

<sup>&</sup>lt;sup>4</sup>Amongst immigrant entrepreneurs from former Yugoslavia, Romania and Poland between 44 and 49 percent have used bank loans as start-up funding, while the same number for entrepreneurs from Turkey, Africa, Iran, Lebanon and Chile is between 31 and 36 percent. Only 26 percent of entrepreneurs from the Middle East seem to have used bank loans when starting up their business.

information used by both the entrepreneur and the bank officials and provide the trust or contacts needed for a positive credit decision. But they might also make relatively favorable decisions towards members of their own group, what is usually referred to as active representation (Bradbury and Kellough, 2011).<sup>5</sup>

Notably, the average share of the bank sector which the own group constitutes is around 0.1%, which indicates that there would be virtually no chance to meet with a co-ethnic bank employee when applying for a loan if the choice of bank office was random. But we do not expect the applicants to randomly approach a bank, instead they are likely turn to the bank where the likelihood of a positive credit decision is the largest.

If there is local representation of the own ethnic group in a local bank office the applicant is expected to turn to the bank where this co-ethnic banker is employed. The likelihood of a positive credit decision at this particular bank may be larger than in other banks regardless if the applicant meets with the own-group employee or not. The reason for this is that the presence of a minority banker may alter the perception of the particular group among other clerks and thereby affect the decisions made by the whole office (Bradbury and Kellough, 2011).

Furthermore, the information that the potential entrepreneurs have regarding the financial system may be improved by residing nearby a banker from the own country. Once there is someone in the group working in a bank at the local level the community has better access to proper information regarding financial businesses. Again, as co-ethnic bankers are very rare, an individual would most likely not benefit from this knowledge without actively searching for it within the community. Therefore the relevant factor is whether the knowledge regarding banking is available in the community, rather than the number of individuals who will be able to provide this knowledge.

As we expect individuals to act non-randomly when choosing bank office, and when searching for financial information, the explanatory variable of interest will be defined as a dummy variable in the empirical model (see section 3.1). This will indicate whether or not an individual is exposed to at least one co-ethnic banker locally. This variable will capture both the probability of improved knowledge, and own-group representation, in the local bank, and the improved financial system knowledge within the local group. Thus we will not be able to distinguish between

<sup>&</sup>lt;sup>5</sup>It might also be the case that own-group bank clerks behave the opposite way and reject applications from their own group in fear of being accused of favoritism. Both the positive and negative effect of active representation are likely of very small importance here since the number of own-group bank clerks is small.

these mechanisms.

As the banks might act upon a local demand from the entrepreneurs through employing individuals from particular countries, we cannot interpret a positive correlation as a causal effect of local co-ethnic bank employees on the probability of being self-employed. Instead a positive OLS estimate should be seen as a market outcome where banks and entrepreneurs act in response to one another. In this process the shared background would function as a mediator for reducing uncertainty and providing information related to credit access.

## 3 Data and sample selection

For this study I use register data over the Swedish labor market for the years 1987 to 2003, where individuals are linked to their workplace through tax records on annual income. In this data I can see if anyone is self-employed.<sup>6</sup> I also have information of the industry of each firm or workplace, from which I can define those working in the bank sector.<sup>7</sup> The data contains information on municipality of residence as well as municipality of workplace, which means that I can study co-ethnic bank employees both residing and working locally.

Apart from this, I also have information on individual characteristics, including country of origin and the immigration year, as well as year of birth, sex, and level of education.<sup>8</sup> I restrict the analysis to the group of non-western immigrants in Sweden, since this group highly overlaps with the group of refugee immigrants for which I can use the placement policy as an source of exogenous settlement. Owngroup individuals is quite broadly defined (17 groups) as small source countries are grouped due to confidentiality reasons.<sup>9</sup> Throughout the text I will use the term co-ethnic or own-group individual as a short term for an individual from the same country/region of origin.<sup>10</sup>

<sup>&</sup>lt;sup>6</sup>Until 1993 self-employment only included sole traders but from 1993 there is information on both sole traders and limited liability companies. For a further description of the definition of self-employment, see the variable description and figure 3 in the appendix.

<sup>&</sup>lt;sup>7</sup>Industry coding for 1987 and 1988 are imputed from the industry coding in 1989 due to missing information in the earlier years. See graphical explanation of this in the variable description in the appendix.

<sup>&</sup>lt;sup>8</sup>For the individuals immigration before 1990 many have missing educational information the first years, so for everyone for whom there is no data indication of further education acquired in Sweden I impute the early observations with the first observed non-missing educational level.

<sup>&</sup>lt;sup>9</sup>See country of origin definitions in variable description in the appendix, and distribution over countries of origin in table 9 in the appendix.

<sup>&</sup>lt;sup>10</sup>I am aware that the crude country of origin measure that is used here does not by any means

I study individuals who were between 18 and 55 when arriving to Sweden, and who arrived between 1987 and 1991 when the placement policy was most effective. I further restrict the sample to individuals who did not have an adult relative already residing in Sweden at the time of arrival, since tied movers were not subject to the governmental placement.<sup>11</sup> This will be referred to as the main sample, and in an extension, I reestimate the model using a larger set of non-western immigrants.

As outcome variable I use a dummy variable capturing self-employment status at a particular point in time, based on individuals for whom self-employment is the main income source.<sup>12</sup> Both individuals who are unemployed and employed in regular employment are thus defined as not being self-employed. In an extension I also study entry into, and exits from, self-employment.

#### 3.1 Defining the explanatory variable

It follows from the discussion in section 2.2 that the processes determining the access to financial information and the choice of local bank office is likely to be non-random. This means that if there is a local own-group bank representative the applicant might turn to the bank where she is employed. Therefore the key explanatory variable is defined as a dummy, *co-ethnic banker*, indicating the exposure to at least one coethnic banker locally.

From the data, the co-ethnic banker dummy can be defined based on the bankers' municipality of residence, or municipality of work. These alternatives would capture slightly different mechanisms, but defining the variable *co-ethnic banker* based on where bankers live or work yields variables which are highly correlated (0.73).<sup>13</sup>

capture an individuals' ethnicity, why the term may be misleading. A similar term which could be used would be co-national, but it also has its limitations. It should be noted that, regardless which concept is used, the variable captures relations between individuals who share the same country/region of origin.

<sup>&</sup>lt;sup>11</sup>A comparison is made with Immigration Boards numbers on residence permits for refugee of asylum reasons, and for each of the studied immigration years my data includes a slightly smaller number than those of the immigration board. My sample only includes individuals arriving at an age between 18 and 55, while their numbers include all immigrants. Another comparison is made with Åslund and Fredriksson (2009) who use the same placement policy for identification. For the immigrations years which they study the sample I use here includes slightly less individuals (20,307 compared to 22,556) and the distribution of countries/regions of origin looks very similar. Therefore I conclude that the data used for instrumental variable analysis here is a reasonable approximation of refugee immigrants entering during the studied years.

 $<sup>^{12}</sup>$ An individual is only coded as self-employed if annual incomes from self-employment is positive.

<sup>&</sup>lt;sup>13</sup>The variables differ mostly in the metropolitan regions, where commuting over municipality borders are common.

This means that the variable will capture both the possibility of meeting a co-ethnic banker in the community and the possibility of meeting a co-ethnic banker in the local bank, whichever of these definitions is used. For this reason I will perform the baseline analysis on the variable based on the municipality of residence of the bankers, and use the variable based on bankers' municipality of work as a sensitivity check.<sup>14</sup>

I also perform a robustness check where the explanatory variable is instead defined as the share of the local co-ethnics who are employed in banks.

#### **3.2** Descriptive statistics

Table 1 shows descriptive statistics for the data containing all non-western immigrants arriving to Sweden between 1987 and 1991, for which I can use the government placement as a source of exogenous settlement. Mean values are also presented divided by treatment status; i.e the value of the *co-ethnic banker* variable.

The data consists of a large share of men, due to the exclusion of tied movers. The studied group consists of both low and highly educated individuals. About half of the individuals in the data do not have a positive annual income, and the earnings are low.<sup>15</sup>

Among the studied individuals, 4 percent are self-employed, and the likelihood of self-employment is increasing with time spent in Sweden (see figure 4 in the appendix). The share employed in a bank is about 0.1 percent.<sup>16</sup> Despite the very low share employed in banks, about 46 percent of the individuals have someone from their own country/region of origin residing in the same municipality who works in a bank. This can partly be explained by the fact that the bankers primarily work in larger municipalities, and municipalities where the group constitutes a larger share

<sup>&</sup>lt;sup>14</sup>In commuting regions bank customers might turn to a bank in a neighboring municipality when applying for a loan. If this is the case, it is not certain what the explanatory variable based on bankers' municipality of work captures. With a co-ethnic banker residing in the municipality, an individual might turn to the bank where this banker is employed, even if that bank is located in a neighboring municipality. Therefore defining the dummy by bankers who reside locally is the preferred choice.

<sup>&</sup>lt;sup>15</sup>Description is presented for the year 1996, which is chosen because it is in the middle of the studied period. For most variables the mean values in 1996 is very similar to the mean values in the pooled data, but for the indicator for having positive annual earnings and mean earnings it is clear that 1996 represents a year with very low employment rate and earnings. This has to do with the economic crisis in Sweden in the middle of the 1990s.

<sup>&</sup>lt;sup>16</sup>When calculating the exposure to co-ethnic bankers a sample consisting of all immigrants from non-western countries is used, regardless of their year of arrival or whether they arrived as tied movers.

	All	Co-ethnic	Co-ethnic	
		banker=1	banker=0	
Age	37.3991	37.2377	37.5373	
Share male	0.5829	0.5793	0.5860	
Education Missing	0.0124	0.0125	0.0124	
No more than 8 years	0.1895	0.1678	0.2080	
9 years	0.1854	0.1800	0.1900	
2 year high school	0.1548	0.1558	0.1539	
3 year high school	0.2024	0.2174	0.1896	
University, at most 2 years	0.1339	0.1405	0.1283	
University, more than 2 years	0.1099	0.1126	0.1075	
PhD	0.0117	0.0134	0.0102	
Positive annual earnings	0.5291	0.5418	0.5182	
Earnings $(1,000 \text{ SEK})$	20.0761	20.2303	19.9442	
Self-employed	0.0400	0.0369	0.0426	
Employed in bank	0.0010	0.0017	0.0005	
Own-group banker	0.4612	1.0000	0.0000	
Placed in metropolitan area	0.4858	0.6338	0.3590	
Municipality size (in 1,000s)	149.4033	251.6320	61.8947	
Municipal group level covarariat	es			
Employment rate	0.4959	0.5187	0.4763	
Share self-employed	0.0352	0.0364	0.0342	
Mean earnings $(1,000 \text{ SEK})$	38.9110	40.9626	37.1548	
Share with university educa- tion	0.2422	0.2601	0.2269	
Share of local population	0.0106	0.0142	0.0076	
Share with above median income	0.2650	0.2797	0.2524	
Mean years in country	9.6549	10.5599	8.8803	
N	53,518	24,683	28,835	

Table 1: Mean individual characteristics year 1996

*Note:* Earnings are the mean annual income from work with a zero income for those without income. Deflated to 1980 values. Mean earnings at the group local level is among individuals with a positive annual income. Distribution over countries of origin presented in table 9 in the appendix. The remaining municipal group level covariates are described in table 10 in the appendix.

of the local population. This means that many countrymen are exposed to each banker's presence. 63 percent of those treated were placed in a metropolitan region at arrival to Sweden and this is true for only 36 percent among those non-treated. The treated group resides in municipalities where the own group consists of more highly educated individuals with higher annual incomes.

The appendix shows mean values for the group municipality level covariates which are not shown here. It also shows the distribution over country of origin. The largest groups in the data are individuals from states in the Middle East, Iran, Chile and Eastern Africa.

### 4 Econometric strategy

#### 4.1 OLS specification

I estimate the effect of the presence at least one local co-ethnic banker on the self-employment probability. Here *i* is the index for individual, *c* is country of origin, *m* is municipality of residence and *t* is year of observation. I also control for individual characteristics such as level of education when arriving to Sweden, sex and age  $(X_{it})$ . When controlling for both country of origin fixed effects  $(\theta_c)$  and municipality fixed effects  $(\psi_m)$  the variation used for identification is the variation within groups between municipalities and within municipalities between groups. I also control for year of immigration fixed effects  $(\rho_{t_0})$ , eliminating the bias from different entry cohorts meeting different labor markets at the time of entry, and year fixed effects  $(\pi_t)$ . A control for time-varying municipality size is also added. Standard errors are clustered at the level of the local group, as this is where the main identifying variation comes from.

$$pr(\text{Self-employed})_{it} = \alpha + \beta \text{Co-ethnic banker}_{it} + \gamma X_{it} + \mu V_{cmt} + \theta_c + \psi_m + \rho_{to} + \pi_t + \epsilon_{it}$$
(1)

This analysis hinges on the assumption that local financial market knowledge and banks in the local area matters for credit decisions. The Agency for Economic and Regional Growth (2012) shows that entrepreneurs in remote and low populationdensity regions perceive capital access as a larger difficulty than entrepreneurs in larger cities, which implies that capital is to some extent supplied at the local level. This supports the claim that local banking is important for the capital acquisition for small firms. In the baseline analysis I focus on the probability of being self-employed at a particular point in time. I will also explore the dynamics of the self-employment entry and exit in an extension.

#### 4.2 Municipal group level covariates

It is possible that *co-ethnic banker* will capture something more than the pure effect of improved access to banks and financial knowledge on self-employment probability. This would be the case if there is some omitted variable which is correlated both to the self-employment rate and the representation of the group in the banks, for example the groups' local labor market position. To rule out this possibility I control for an extensive set of covariates at the level of the local group  $(V_{cmt})$ .<sup>17</sup>

A number of variables are included to control for the labor market position of the group at the local level, such as the employment rate, mean earnings and share with income above the median. I also control for share with university education. But as these variables might be imperfect measures of how well the group is integrated at the local labor market, I also control for the mean income of, and share of university educated at, the workplaces where the co-ethnics of an individual work.

It should be noted here that even though ethnic businesses only constitute a small fraction of immigrant businesses in Sweden (Committee on Foreign Born Entrepreneurship, 1999) it is likely that ethnic businesses will start up as a response to a demand for ethnic goods. A high share of the co-ethnics working in the formal bank sector might imply high purchasing power of the group, which could possibly be correlated with a high demand for ethnic products. If this is the case, we could see that a high share of co-ethnics in the bank sector positively correlates with the probability of becoming self-employed due to the omitted demand variable. But this risk is reduced by the inclusion of group local level income variables, as discussed above. The demand for ethnic goods is also potentially larger among more recent immigrants, which means that another way to control for ethnic demand is to include controls for the mean time spent in Sweden in the local community.

As banks might employ clerks from a specific country in response to the credit demand from this specific group it is useful to control for the share of the individuals from ones' own country who are themselves self-employed. The own-group members defining this variable includes all immigrants from the same country in the munic-

<sup>&</sup>lt;sup>17</sup>These variables are, just as the *co-ethnic banker* variable, defined in a sample consisting of all immigrants from non-western countries, regardless of their year of arrival or whether they arrived as tied movers. The variables are aggregated at the level of the own group in the municipality of residence, after excluding the own observation.

ipality, except for the individuals in the main sample. The individuals in the main sample are excluded as this control variable otherwise will be a group municipality level aggregation of the outcome variable.<sup>18</sup> The group might also be represented in the local bank only because the group constitutes a large share of the population in the municipality. For this reason I control for the share of the population which the own group constitutes at the local level. For a full description of the covariates included, see variable list and table 10 in the appendix.

## 5 OLS results

Table 2 presents the baseline OLS estimates for the probability of being self-employed in a given time period t. The estimate for the whole group is presented in column 1, and the remaining columns presents results for different subsamples. Column 1 shows that having a local co-ethnic working in a bank is associated with a 0.6 percentage point increase in the probability of being self-employed which translates to about a 15% increase in the mean self-employment rate. In this study, both individuals who are unemployed and have a regular employment are coded as not being self-employed, why an increasing self-employment rate should not be interpreted as an increased employment rate in the group.

We see that for those individuals who arrived to Sweden with at most high school education (column 2 and 3) co-ethnic bank employees seem to matter for the selfemployment probability. For the least educated the size of the estimate indicates that the presence of at least one co-ethnic local bank employee is associated with an increase in the self-employment rate of about one fifth of the mean. Similar effects are found for males, while there are no indications of a positive relation for females. The mean self-employment rate among women is 1.7%, and thus much smaller than that of men (5.3%). The variables included in the model also explain a smaller share of the variation in female self-employment rates than it does for men. This indicates that the processes determining male and female self-employment differ and

<sup>&</sup>lt;sup>18</sup>The results are robust to both excluding this control variable, and to defining it by all co-ethnics (except for the own observation, as with the other group municipality level variables). As this control variable controls away for one possible mechanism through which co-ethnic bankers might affect an individuals' self-employment probability, namely through the groups' entrepreneurial activity, the least conservative estimates are attained if not including this control at all. The most conservative estimates are obtained when aggregating self-employment over all local co-ethnics. To avoid that the control variable captures the same thing as the outcome variable, my preferred choice is to include this control variable, but to define if by the individuals from the own group who are not themselves in the main sample.

observed individual and group-municipality level factors are poorer determinants of female self-employment.

For individuals with more than high school education, the estimates are small and insignificant. The substantially smaller effect of local co-ethnic bankers on the self-employment probability for the highly educated can potentially be explained by the fact that this group might have better observable characteristics, and aim to start businesses in less over-crowded sectors. Both of these factors may function as signals of high productivity, and will thereby affect the banks' capital allocation decisions even in absence of the help from co-ethnic bank employees.

But the estimate for the highly educated might also be affected by the fact that bank employees do not only facilitate credit access. They may also offer labor market contacts which are beneficial for entry into the regular labor market, and this would counteract the positive effect on the likelihood of being self-employed. As individuals with higher education are more likely to share labor market with the bankers than less educated individuals are, this labor market referral effect is more likely to influence the estimates for this group.

The last two columns show estimates for the individuals who reside in metropolitan and non-metropolitan regions separately.<sup>19</sup> The relation between co-ethnic bankers and self-employment rate is twice as large in the non-metropolitan region as in the metropolitan region.

<sup>&</sup>lt;sup>19</sup>Actually, for facilitation of comparison with estimates using the initial year exogenous variation, these subsamples are defined by the municipality at time of arrival. Metropolitan regions are here defined as Stockholm county, Västra Götaland country (the region where the second largest city Gothenburg is located) and Scania county (the region where the third largest city Malmö is located).

	(1) All	(1) Less than high school	(2) High school	(3) More than high school	(4) Men	(5) Women	(6) Metro	(7) Non-metro
Co-ethnic banker	$0.00580^{***}$ (0.00148)	$0.00840^{***}$ (0.00188)	$0.00554^{***}$ (0.00185)	0.000160 (0.00253)	$0.00912^{***}$ (0.00203)	0.000937 (0.00122)	$0.00361^{**}$ (0.00161)	$0.00700^{***}$ (0.00213)
Mean outcome	0.038	0.036	0.04	0.039	0.053	0.017	0.035	0.04
Observations R-squared	800,899 0.053	325,430 $0.067$	303,818 0.053	$171,651 \\ 0.049$	467,226 0.063	333,673 $0.024$	390,613 0.056	410,286 0.057
Note: The table rep	orts estimated $\beta$ <sub>F</sub>	parameters (standard e	errors) from equat	tion (1). The outcom	ie variable is a dum	umy for being self-e	employed. Robust st	andard errors are

Table 2: Estimated effects of exposure to co-ethnic banker on self-employment probability: OLS

clustered at the level of each municipality country of origin pair. \*\*\* p<0.01, \*\* p<0.05, \* p<0.1. Fixed effects for year-, year of immigration-, municipality- and country of origin are included as well as controls for sex, age and education level. A full set of municipal group level covariates are also included, see list in the appendix for overview of these variables and table 14 in the appendix for full regression output (for columns 1 and 2).

## 6 Analysis based on initial exogenous variation

As discussed earlier the direction of causality of the OLS estimates is not clear, as banks and entrepreneurs act in response to one another. Individuals may also select into municipalities where the group's labor market position is favorable. As the presence of co-ethnic bankers is related to slightly better labor market outcome of the group overall (see table 1), this selective migration may bias the estimates. But it is not clear which way the bias from this selective migration will affect the estimates, as own group individuals in the bank sector may be associated with larger possibilities for employment in the regular labor market as well as facilitated credit access. The bias arising from the fact that banks might act in response to entrepreneurs' demands is most likely affecting the estimates positively.

To get closer at a causal understanding of the exposure to co-ethnic bankers and its' effect on self-employment probability I make use of a refugee dispersal policy, which provides arguably exogenous variation in the exposure to co-ethnic bankers the first year in the country. By studying initial year exposure we can rid the estimates of the potential bias arising from individuals' settlement patterns.

#### 6.1 Placement of refugees: the policy

Between 1985 and 1994 refugees arriving to Sweden were subject to a dispersal policy through which the government was responsible for placing them in municipalities which met certain standards. At first, a smaller share of the municipalities were involved. But due to a substantially increased inflow of refugees the years following 1985, the number of municipalities contracted to receive refugees increased and by 1987 almost all Swedish municipalities were involved (Borevi and Myrberg, 2010; Andersson et al., 2003). The placement was based on agreements made with the local municipalities and was designed to disperse the refugees from the metropolitan areas over a larger number of municipalities in Sweden (Borevi and Myrberg, 2010).

The refugees themselves could ask for a residential location, but due to scarcity of available housing in many municipalities the wishes could most often not be fulfilled. During the years 1987 to 1991 the immigration numbers were large and thus the likelihood of being assigned to the preferred municipality was substantially reduced compared to earlier and later years. A reason for this was that priority was given to shortening the time span between receiving residence permit and moving to the placement municipality, rather than to accommodating to the individuals' preferences (The Immigration Board, 1997).

The officials in charge for placement did not meet with the refugees, but based

the placement on information from the refugee centers (The Immigration Board, 1997). This means that placement was based primarily on observable factors, and thus we can arguably control for the factors which might have affected placement (Åslund and Fredriksson, 2009).

Many researchers have shown that the dispersal policy did indeed change the initial settlement patterns among the refugees (The Committee on Immigration Policy, 1996; Andersson, 1998). Prior to this reform about 7% of the inflow of non-OECD immigrants settled in the six northernmost counties of Sweden, and by the end of the policy period this fraction was 14 % (Åslund et al., 2003). During the same period the fraction of the refugees settling in Stockholm decreased.

After initial placement, there was nothing keeping the immigrants from onward migration to a municipality which they preferred better, and non-trivial mobility rates indicate that many did not settle with the municipality of placement (Andersson et al., 2003). Åslund et al. (2010) find that the immigrants who arrived during this period started out in localities which differed from those of previous immigrants, but over time the characteristics of the localities of residence became more similar. Similarly, Andersson (1998) shows that most secondary migration has been into cities with a larger concentration of the own ethnic group than the reception municipality. Despite these onward migrations rates around 56 percent of the immigrants remained in the municipality of initial placement after 5 years in the country.

The above description indicates that refugees arriving to Sweden between 1987 and 1991 were restricted in their first location of residence. Thus, conditional on observed covariates, the assignment of placement municipality can be treated as exogenous to the outcome.

#### 6.2 Modeling the impact of placement

Due to the possibility of controlling for observable characteristics of the refugees, the policy described above can be used to provide exogenous settlement municipalities for the refugees placed during this period. This means that the exposure to co-ethnic bankers the first year in Sweden is arguably exogenous, conditional on observed characteristics, and we can model self-employment probabilities at later points in time as a function of this initial exposure. This rids the estimates of potential bias from residential sorting. Also, when modeling the self-employment probability as a function of earlier years' exposure to co-ethnic bankers, the potential bias due to reverse causality is reduced. The estimate from this specification can be interpreted as the effect of initial exposure to own-group bank employees on the self-employment rate, and it should therefore be seen as a long term effect. Standard errors are here clustered at the arrival municipality group level, and the initial characteristics of the group at the local level are controlled for, instead of the contemporary characteristics.<sup>20</sup> Also municipality of arrival fixed effects are used instead of fixed effects for the contemporary municipality.

Self-employed<sub>*it*</sub> =
$$\beta$$
Co-ethnic banker<sub>*it*0</sub> +  $\gamma X_{it}$  +  $\mu V_{cmt0}$  +  $\theta_c$  +  $\psi_{mt0}$   
+  $\rho_{t0}$  +  $\pi_t$  +  $\epsilon_{it}$  (2)

For this model to causally estimate the impact of initial year exposure on subsequent self-employment probability, the initial placement has to be randomly assigned, and independent of the potential outcome. As a test of this I predict the propensity to become self-employed based on observable individual level covariates and study the correlation between this propensity and *co-ethnic banker* variable in the arrival municipality the arrival year. This correlation is -0.0498, which implies that those with characteristics associated with a higher propensity to become selfemployed are not to a larger degree placed in municipalities where co-ethnic bankers are present than where they are not present. This supports the claim that the settlement patterns is exogenous relative to the ethnic composition in the financial sector.

In contrast to the IV-version of equation (1) (described below), specification (2) can be seen as the direct effect of the instrument on the outcome. From here on I will therefore refer to specification (2) as the reduced form model.

#### 6.3 Instrumental variable analysis

Another way to use the exogenous first year variation in exposure to co-ethnic bankers is as an instrument for the endogenous regressor in the OLS regression of equation (1). This means using the *co-ethnic banker* variable in the arrival municipality the arrival year as instrument for the same variable in the municipality of residence the year of interest. Equation (3) shows the first stage regression for

<sup>&</sup>lt;sup>20</sup>As a sensitivity test, I test whether the results are affected by defining these variables only by the individuals who were residing in the arrival municipality at the year of arrival. This means that the individuals who arrived the same year as yourself are not included in the aggregate local group level variables. The results are robust to this restriction.

the potentially endogenous variable *co-ethnic banker*. The predicted contemporary co-ethnic banker variable is used as explanatory variable in the second stage.

Co-ethnic banker<sub>it</sub> = 
$$\beta$$
Co-ethnic banker<sub>it0</sub> +  $\gamma X_{it}$  +  $\mu V_{cmt_0}$  +  $\theta_c$  +  $\psi_{mt_0}$   
+  $\rho_{t_0}$  +  $\pi_t$  +  $\epsilon_{it}$  (3)

For the co-ethnic banker dummy in the arrival municipality to be a valid instrument, it must be correlated with the same variable in the municipality of residence the year of interest, conditional on the other covariates. This assumption is testable and results are presented in a first stage analysis. As the strength of the instrument decreases with time spent in the country, this assumption will lose its validity the longer the individual has been in the country.

Furthermore, for the instrumental variable analysis to be valid, the exclusion restriction has to be fulfilled. This means that the co-ethnic bank employees in the arrival municipality the arrival year cannot affect the probability of being selfemployed in the year of interest in any other way than through the presence of a co-ethnic bank employee in year t. As co-ethnic bankers the arrival year might be related to own-group bank representation in years between the arrival year and year t, this assumption might not be fulfilled. This could affect self-employment probability in year t through the start-up of a business any of the earlier years.

The potential invalidity of the exclusion restriction makes it difficult to interpret the size of the instrumental variable estimate. Despite this, I will present these estimates alongside estimates from specification (2).

#### 6.4 Results based on initial exogenous variation

Table 3 presents results from estimations using the exogenous initial exposure to co-ethnic bankers as explanatory variable. Here the first panel shows the reduced form estimates of the effect of the co-ethnic bankers in the placement municipality on the contemporary self-employment rate. Panel B shows the IV estimate where the co-ethnic banker variable in the arrival municipality year  $t_0$  is used to instrument the same variable in the municipality of residence the observation year. Panel C presents the first stage estimates, which are large and highly significant throughout.

Here results for the whole sample and for the subsamples for which we saw significantly positive OLS estimates are presented. For individuals with less than high school education, males and those placed in a metropolitan region the estimates are in line with earlier OLS estimates. For these groups we see significantly positive point estimates for the reduced form and/or the IV. For those with high school education and those placed in a non-metropolitan region, the OLS estimates were significantly positive, but IV and reduced form estimates are insignificant. This can be seen as indications of that the OLS estimates are driven by residential sorting, and when taking this into account, there is no clear effect of co-ethnic bankers. The slightly smaller reduced form than OLS estimates for the low educated and for the males might be indications of that residential sorting influences the OLS estimates also for these groups. Results for the other subsamples are found in table 12 in the appendix (for these subsamples we see insignificant reduced form and IV estimates).

The placement of refugees over the Swedish municipalities has been used extensively for evaluating peer effects and effects of local community composition on later labor market outcomes. One general finding of these studies is that sorting between localities matters substantially for the conclusions of how the size of the ethnic community affects subsequent labor market and schooling outcomes (Åslund and Fredriksson, 2009; Åslund et al., 2011, 2003). When not accounting for residential sorting, the estimates generally point towards worse schooling and labor market outcomes as a consequence of a high concentration of the own-group, while when taking this sorting into account, the opposite conclusion is reached.

The results of this study are not directly comparable to the above mentioned studies, as the exposure to a co-ethnic banker is not a direct measure of the quantity of the local group. The concentration of the local group thus seems to be a factor relevant in the residential mobility decision of immigrants, while it is less likely that residential decisions are made on the basis of the presence of an own-group banker. This might be a reason why residential sorting does not seem to affect these results as much as it does affect results from earlier studies using the same placement policy.

As discussed earlier, the size of the IV estimates may be affected by the fact that the exclusion restriction might not be fully valid. There is reason to believe the bank employees in years in between the arrival year and year t has been influential in the decision to become self-employed, something which could possibly influence the IV estimate. The reduced form estimates tell us that, for low educated, males and those placed in metropolitan region (insignificant) there is a long term effect on self-employment probability of the exposure to local co-ethnic bankers at arrival to Sweden. The similar magnitudes of the OLS and reduced form estimates may be interpreted as an indication of that identification does not rely on individual variation over time, but rather on variation in the cross-section.

			(1)		(4)	(c)
	All	Less than high	High school	Men	Metro	Non-metro
		scnool				
Panel A. Reduced form						
Co-ethnic banker , year $t_0$	0.00207	$0.00562^{*}$	1.88e-05	$0.00576^{*}$	0.00283	0.000189
	(0.00186)	(0.00294)	(0.00291)	(0.00318)	(0.00244)	(0.00301)
Observations	800,899	325,430	303,818	467,226	390,613	410,286
R-squared	0.047	0.061	0.047	0.055	0.045	0.049
Panel B. IV						
Co-ethnic banker year t	0.00845	$0.0219^{*}$	8.27e-05	$0.0246^{*}$	$0.0225^{***}$	0.000755
	(0.00754)	(0.0113)	(0.0128)	(0.0135)	(0.00842)	(0.0120)
Observations	800,899	325,430	303,818	467,226	390,613	410,286
R-squared	0.046	0.057	0.047	0.052	0.041	0.049
Panel C. First stage						
Co-ethnic banker, year $t_0$	$0.246^{***}$	$0.256^{***}$	$0.227^{***}$	$0.234^{***}$	$0.235^{***}$	$0.250^{***}$
	(0.0193)	(0.0210)	(0.0212)	(0.0192)	(0.0232)	(0.0302)
Observations	800,899	325, 430	303,818	467, 226	390,613	410,286
R-squared	0.315	0.344	0.297	0.297	0.376	0.186

as well as controls for sex, age and education level. A full set of municipal of arrival group level covariates are also included, see list in the appendix for overview of these

variables.

Table 3: Estimated effects of exposure to co-ethnic banker on self-employment probability: instrumental variable and reduced

#### 6.5 The importance of local group quality

The literature using the dispersal of refugees over Sweden to study effects of community characteristics on labor market outcomes suggest that the characteristics of group in the placement municipality affects long run labor market outcomes for the placed immigrants (see for example Åslund and Fredriksson (2009) and Åslund et al. (2011)). Åslund et al. (2003) show that the quality of the initial placement community, measured by either group annual earnings or group self-employment rates, is an important factor for determining labor market outcomes in the long run, independent on educational background. For this reason it is important to control for a range of municipal-group level characteristics of the initial placement to be sure that the co-ethnic banker dummy is not capturing other dimensions of the group's attachment to the local labor market.

Table 4 describes how the reduced form estimate changes with the inclusion of placement municipality group level covariates. Results are shown for the whole placement data as well as for only those with the lowest educational attainment, since this is one of the groups for which we see the strongest relation. The first column shows estimates when only controlling for individual level characteristics, as well as municipal level-, year-, year of immigration- and country of origin fixed effects. Column 2 adds controls for the municipality size and the share of the local population which the group constitutes, and this reduces the estimate by 0.1 percentage point for the full sample as well as for the low educated. For the full sample the estimate becomes insignificant.

When in the following columns adding controls for the share of the individuals from the own country who are themselves self-employed, and several other group municipality level covariates the estimates are relatively stable. This means that the co-ethnic banker dummy is not highly correlated with other group municipality level covariates related to the labor market position and the quality of the group at the local level. I therefore draw the conclusion that the estimated effect is not driven by characteristics of the group in the initial location.

## 7 Extensions

#### 7.1 Own-group employees of a similar sector

As a test of whether the explanatory variable of interest captures more than the pure capital access effect we are interested in, I reestimate the baseline model adding a control for own-group employees in a similar sector. The sector that I study is the

(1)	(2)	(3)	(4)	(5)
$0.00314^{*}$	0.00221	0.00208	0.00216	0.00207
(0.00186)	(0.00189)	(0.00187)	(0.00186)	(0.00186)
000 000	000 000	000 000	000 000	000 000
800,899	800,899	800,899	800,899	800,899
0.046	0.046	0.047	0.047	0.047
$0.00684^{**}$	$0.00590^{**}$	$0.00556^{*}$	$0.00558^{*}$	$0.00562^{*}$
(0.00277)	(0.00297)	(0.00294)	(0.00293)	(0.00294)
205 420	205 420	205 420	205 420	205 420
325,450	525,450	325,450	325,450	323,430
0.060	0.060	0.060	0.061	0.061
Yes	Yes	Yes	Yes	Yes
	Yes	Yes	Yes	Yes
		Yes	Yes	Yes
			Yes	Yes
				Voc
	(1) 0.00314* (0.00186) 800,899 0.046 0.00684** (0.00277) 325,430 0.060 Yes	(1)     (2)       0.00314*     0.00221       (0.00186)     (0.00189)       800,899     800,899       0.046     0.045       0.00684**     0.00590**       (0.00277)     325,430       0.060     325,430       0.060     Yes       Yes     Yes	(1)       (2)       (3)         0.00314*       0.00221       0.00208         (0.00186)       (0.00189)       (0.00187)         800,899       800,899       800,899         0.046       0.046       0.047         0.00684**       0.00590**       0.00556*         (0.00277)       0.00590*       0.00594         325,430       325,430       325,430         0.060       0.060       0.060         Yes       Yes       Yes         Yes       Yes       Yes         Yes       Yes       Yes         Yes       Yes       Yes         Yes       Yes       Yes	$(1)$ $(2)$ $(3)$ $(4)$ $0.00314^*$ $0.00221$ $0.00208$ $0.00216$ $(0.00186)$ $(0.00189)$ $(0.00187)$ $(0.00186)$ $800,899$ $800,899$ $800,899$ $800,899$ $0.046$ $0.046$ $0.047$ $0.047$ $0.00684^{**}$ $0.00590^{**}$ $0.00556^*$ $0.00558^*$ $(0.00277)$ $0.00590^{**}$ $0.00556^*$ $0.00558^*$ $325,430$ $325,430$ $325,430$ $325,430$ $0.060$ $0.600$ $0.060$ $0.061$ Yes

Table 4: Estimated effects of exposure to co-ethnic banker on self-employment probability: reduced form with different sets of covariates

Note: The table reports  $\beta$  parameters (standard errors) from equation (2). The outcome variable is a dummy for being self-employed. Standard errors are clustered at the level of municipality of arrival and country of origin pairs.\*\*\* p<0.01, \*\* p<0.05, \* p<0.1 Fixed effects for year-, year of immigration-, municipality of arrival- and country of origin are included in all columns, as well as controls for sex, age and education level. Column 2 adds controls for the municipality size and the share of the local population which the group constitute. Column 3 then adds a control for the share of the group at the local level who are self-employed. Column 4 then adds further controls for the earnings and education level of the group at the local level, while column 5 also adds controls for characteristics of the workplaces where the own-group members locally are working. See list in the appendix for overview of these variables. insurance sector. The two sectors are similar in the sense that they require the same amount and type of education and offer similar wages (see table 11 and figure 1 in the appendix). But the insurance market does not provide credit for entrepreneurs. Although we expect insurance company employees to have partial knowledge of the financial sector and potential contacts within it, and hence also potentially influence the capital access process, we expect the effect of co-ethnic bankers to exceed the effect of own-group insurance company employees. As exposure to owngroup employees in these sectors is correlated, we include both variables in the model and study the effect of both simultaneously.<sup>21</sup>

Both banks and insurance companies are sectors in which non-western immigrants earn relatively high incomes and those who are employed have relatively high education.<sup>22</sup> Therefore, employment in any of these sectors can be considered a relatively high level of economic integration for a non-western immigrant. As discussed earlier, bank employees might affect the probability to enter self-employment not only through improved credit access but also through affecting the possibilities of employment in the regular labor market. Insurance sector employees from the own country should mainly capture the second of these effects.

Figure 2 in the appendix shows the total number of non-western immigrants employed in the bank and insurance sector during the time period studied. There are much fewer non-western employees in the insurance sector than in the bank sector, mainly because the insurance sector employs much fewer workers overall. The share of employees with a non-western background is similar in the two sectors. In the analysis I focus on a co-ethnic insurance dummy, defined analogously to the co-ethnic banker dummy. The share of individuals who are exposed to at least one own-group insurance employee is 0.31.

Table 5 presents the results from the comparison between the local co-ethnic bank and insurance sector employees, focusing on the whole group and the subsamples for which we found significant results in the previous analysis. I also limit the attention to the OLS and reduced form estimates.

The results show that the patterns from the baseline analysis remains, with positive OLS and reduced form estimates of bank representation on the probability of being self-employed, particularly for low educated and for males. Reassuringly,

<sup>&</sup>lt;sup>21</sup>Municipality-year observations where there is no workplace identified as insurance company are not included in this analysis.

<sup>&</sup>lt;sup>22</sup>Some non-western bank employees have relatively low income. These may be individuals employed at maintenance positions, such as janitors and cleaners. As we cannot distinguish these individuals, this may lead to an underestimation of the effect of bank employees who are involved in the credit access process.

	(1)		
Bonol A All	(1)	(2) Reduced form	
ranei A. All	OLS	Reduced form	
Co-ethnic banker	$0.00570^{***}$	0.00201	
	(0.00144)	(0.00186)	
Co-ethnic insurance employee	0.00144	-0.00412	
	(0.00172)	(0.00256)	
Observations	800.899	800.899	
R-squared	0.053	0.047	
a to A to the state			
Panel B. Less than high school	OLS	Reduced form	
Co-ethnic banker	0.00832***	0.00556*	
	(0.00188)	(0.00294)	
Co-ethnic insurance employee	0.00113	-0.00714**	
	(0.00238)	(0.00328)	
Observations	$325,\!430$	325,430	
R-squared	0.067	0.061	
Panel C. Male	OLS	Reduced form	
Co-ethnic banker	0.00889***	0.00557*	
	(0.00197)	(0.00319)	
Co-ethnic insurance employee	0.00308	-0.0104**	
	(0.00244)	(0.00407)	
Observations	467.226	467.226	
R-squared	0.063	0.055	
<b>Banal D</b> Matropolitar region	OIS	Reduced form	
ranei D. Metropontan region	ULS	Reduced Iorin	
Co-ethnic banker	$0.00361^{**}$	0.00249	
	(0.00158)	(0.00242)	
Co-ethnic insurance employee	6.26e-05	-0.00697**	
	(0.00192)	(0.00275)	
Observations	390,613	390.613	
R-squared	0.056	0.045	

Table 5: Estimated effects of exposure to co-ethnic bankers and insurance employees on self-employment probability: OLS and reduced form

Note: Column 1 (2) reports  $\beta$  parameters (standard errors) from a version of equation (1) ( (2)), where both Co-ethnic bankers and own-group insurance company employees are used as explanatory variables. The outcome variable used is a dummy for being self-employed. Standard errors for column 1 (2) are clustered at the level of municipality (of arrival) and country of origin pairs.\*\*\* p<0.01, \*\* p<0.05, \* p<0.1 Fixed effects for year-, year of immigration-, municipality- and country of origin are included, as well as controls for sex, age and education level. In column 1 municipal group level covariates are contemporary, and in column 2 the initial year value is used. In column 2 also municipality of arrival fixed effects are used instead of contemporary municipality fixed effects. See list in appendix for overview over these variables.

we do not find corresponding OLS results for the effect of the co-ethnic insurance dummy. The reduced form estimates are significantly *negative*. For the groups not presented, I generally do not find any significant effects of the insurance sector employees.<sup>23</sup>

The negative estimates for the co-ethnic insurance dummy in the reduced form model may indicate that the insurance company employees capture the impact of a well (economically) integrated group. The exposure to such individuals may be positively related to possibilities of regular employment. The fact that the sign is the opposite to the estimate for co-ethnic bankers suggests that the two professions have very different relations to self-employment, despite the similarity in labor market status. In my view, this strengthens the argument that the co-ethnic banker dummy captures access to capital, and not the impact of having co-ethnics in prestigious labor market segments.

#### 7.2 Dynamics of self-employment: entry and exit

The baseline analysis has been focusing on the stock of self-employed individuals at a particular point in time. This section replaces the outcome variable *Being selfemployed* with the outcomes *Becoming self-employed* and *Leaving self-employment*. These are both defined from the same self-employment variable as before, and the variable for becoming self-employed takes on the value 1 if the individual was not self-employed in the previous year but is self-employed in year t, and 0 if she does not enter self-employment. The variable for leaving self-employment takes on the value 1 if the individual was self-employed the previous year but is not so this year. Similarly, this is set to 0 if she continues to be self-employed.<sup>24</sup>

If access to capital is important both in the start-up and expansion phase and for surviving downturns we could expect to see effects along the extensive margins of both entering and exiting self-employment. Andersson Joona (2010) shows that net wealth at business start-up is an important factor in determining the risk of exiting self-employment and entering unemployment. As is clear from table 6, the local co-ethnic bank employees seem to be related to both self-employment entry and exit. The probability of starting a firm is much lower than the probability of

 $<sup>^{23}</sup>$ For individuals placed outside the metropolitan region we see a significantly positive OLS estimate (significant on the 10 % level).

<sup>&</sup>lt;sup>24</sup>For all observations where an individual was self-employed the previous year, the variable for becoming self-employed is set to missing, as you cannot start up a firm if you are already running one. For observations where an individual was not self-employed the previous year the variable for leaving self-employment is set to missing.

leaving self-employment once having started the firm. The large differences between the magnitude of the estimates for starting up a firm and leaving self-employment are related to the different mean probabilities. Both the effect on entry and the effect on exit is about one tenth of the mean probability, with slightly larger estimates for the low educated. This suggests that local co-ethnic bank employees are related to both facilitated entry into, and reduced exit out of, self-employment.<sup>25</sup>

There are, however, two reasons to be cautious when interpreting these results. As we saw earlier, the effect of initial exposure is almost the same size as the effect of contemporary exposure. This indicates that the effect may materialize with lags, and the transition estimates are identified by contemporary effects. Furthermore, the analysis on the dynamics of self-employment is also performed including insurance sector employees as an explanatory variable. Here we see similar estimates for the insurance sector employees as we do for the bank employees (see table 13 in the appendix).

	А	11	Less than h	nigh school
	(1)	(2)	(3)	(4)
	Entry	Exit	Entry	Exit
Co-ethnic banker	$0.00146^{**}$	-0.0180*	$0.00190^{**}$	-0.0335*
	(0.000642)	(0.00948)	(0.000754)	(0.0189)
Mean outcome	.013	.228	.012	.234
Observations	774,251	26,495	315,157	$10,\!192$
R-squared	0.013	0.041	0.018	0.059

Table 6: Estimated effect of co-ethnic banker on probability of entering and exiting self-employment: OLS

Note: The table reports  $\beta$  parameters (standard errors) from equation (1). The outcome variable used in column 1 and 3 (2 and 4) is a dummy for becoming self-employed (exiting self-employment). Standard errors are clustered at the level of municipality and country of origin pairs.\*\*\* p<0.01, \*\* p<0.05, \* p<0.1 Fixed effects for year-, year of immigration-, municipality- and country of origin are included, as well as controls for sex, age and education level. Municipal group level covariates are contemporary, see list in appendix for overview over these variables.

 $<sup>^{25}</sup>$ IV estimates are in line with these OLS estimates but standard errors are large so they are not significant.

## 8 Sensitivity

Table 7 shows OLS estimates for a wide range of sensitivity checks of the baseline results. Only results for the whole group and for those who arrived with low education are presented.

As noted earlier the local co-ethnic bank employees can be defined based on their municipality of residence or municipality of work. The co-ethnic banker dummy variables based on either of these definitions are highly correlated. As a sensitivity analysis I perform an analysis defining the local co-ethnic bank variable by banker's municipality of work. OLS estimates are presented in column 1 of table 7 and these estimates are very similar in magnitude to the estimates in the baseline analysis.<sup>26</sup>

The second column shows estimates using a variable defined as the *Share of local co-ethnics employed in bank*. The size of these estimates suggest that self-employment probability increases by 0.43 percentage points if the share of the local own-group members who work in a bank increases with one percentage point.

In column 3 I study the full sample of grown up non-western immigrants arriving after 1986. The positive OLS estimate is robust to using this data, which tells us that the individuals in the placement data and the individuals who arrived as tied movers do not differ substantially from each other in this regard. The estimates here are slightly smaller, something which could be due to the larger share of women in this data set.

Column 4 shows estimates when using logit instead of OLS and results are robust to this as well. The log odds of being self-employed, versus not being self-employed, for low educated, increase by a factor of 0.2 if exposed to a co-ethnic banker. This can be translated into that being exposed to a co-ethnic banker increases the odds of being self-employed by 22 %.

<sup>&</sup>lt;sup>26</sup>Using this definition of local co-ethnic employees OLS estimates are more similar for bank employees and insurance company employees. But the estimates for the insurance company employees generally turn significantly negative in the IV and reduced form estimations, as in table 5. This definition is not used as baseline as it is not completely clear what it captures. Using this definition, exposure is mainly taking place in the larger cities, even though it might be the case that residents of commuting municipalities around the cities are also turning to these in-city bankers when applying for loans. This definition of the explanatory variable does not distinguish the effect of bank employees from the effect of employees in similar sectors, and thus it seems to capture the effect of the group's local labor market position rather than the effect of local exposure to co-ethnic bankers.

Panel A. All	(1) Banker dummy by municipality of work	(2) Share	(3) Full data	(4) Logit	(5) Contemporary Education	(6) Sole traders	(7) Firm level code
Co-ethnic banker	0.00527*** (0.00190)		$0.00212^{**}$ (0.000871)	$0.0992^{***}$ (0.0384)	$0.00591^{***}$ (0.00149)	$0.00602^{***}$ (0.00148)	$0.00520^{***}$ (0.00145)
Share of own-group members		$0.439^{**}$ (0.111)	~	~		~	
Observations R-squared	800,899 0.053	800,899 0.049	2,536,065 0.042	798,654	791,278 0.053	801,126 0.050	800,899 0.053
<b>Panel B.</b> Less than high school							
Co-ethnic banker	0.00710*** (0.00201)		$0.00329^{***}$ (0.00123)	$0.196^{***}$ (0.0560)	$0.00914^{***}$ (0.00193)	$0.00857^{***}$ (0.00193)	0.00756*** (0.00182)
Share of own-group members		$0.389^{**}$ $(0.175)$					
Observations R-squared	325,430 0.067	325,430 $0.067$	825,282 0.058	320,677	287,731 0.072	325,550 $0.065$	325,430 $0.067$
<i>Note:</i> The table reports variable based on banker OLS estimation on a sai logistic regression. Colur includes sole traders. Col	estimated $\beta$ parameters (s s' municipality of work, ar nple which is not restrict an 5 use the contemporary umn 7 uses explanatory vi	tandard errors) fron nd column 2 uses th ed to those immigr ~ education level in ariables where banh	n equation (1). The or the share of the own gro ating between 1987-19 stead of the education is are defined by indus'	trcome variable is a c up at the local level 91. This data also level at time of arriv ry code at the firm	lummy for being self-en who are employed in a includes tied movers. "al, and column 6 use a evel instead of at the w	nployed. Column 1 v bank. Column 3 pe Column 4 estimates self-employment de orknlace level. Rob	uses an explanatory reforms the baseline s equation (1) as a finition which only ust standard errors

of origin are included as well as controls for sex, age and education level. A full set of municipal group level covariates are also included, see list in the appendix for overview of these variables.

So far the analysis has relied on the education level at time of arrival to Sweden, but in column 5, I study how the results change when using contemporary education levels, in column 6 I use a self-employment definition only including sole traders, but not limited liability companies.<sup>27</sup> Lastly, column 7 checks to see if the results are robust to using firm level industry codes instead of the workplace level industry codes, which have been used in the baseline analysis. OLS estimates are robust to all these alterations.

Overall, these robustness checks suggest that the baseline results for the full group and for the low educated are robust to a wide range of alterations of how the variables are defined and the way the model is specified.<sup>28</sup>

## 9 Concluding remarks

By studying how the self-employment rate of immigrants is related to exposure to co-ethnic bankers, we derive knowledge about whether shared background can help to overcome the information asymmetries which are associated with capital access for small firms. OLS estimates suggest that exposure to co-ethnic bank employees is positively related to the self-employment probability amongst immigrants, but also that this relationship differs between groups. We see a strong relation for low educated and for males, while we see no indications of positive effects for the highly educated and for women.

The differences between the groups becomes even clearer when employing a placement policy as a source of exogenous settlement to get closer at a causal understanding of the relation. By using this policy we rid the estimates of potential bias from selective residential sorting and reverse causality. The reduced form estimates for low educated, for males and for those placed in a metropolitan regions are in line with the earlier OLS estimates, indicating that there is a positive relation even after controlling for residential sorting. For high school educated and those placed in a non-metropolitan region, reduced form estimates are insignificant while the OLS estimates are significantly positive. This can be understood as signs of sorting into municipalities with own-group bank representation as well as beneficial business

<sup>&</sup>lt;sup>27</sup>Figure 3 in the appendix show the development of the number of individuals in the data who are self-employed over time, using these two different definitions.

<sup>&</sup>lt;sup>28</sup>Some of these estimations are also performed in the reduced form model (results not presented here). For columns 1 and 5 reduced form estimates are significantly positive for the low educated, and for column 2 they are significant also for the full sample. For the remaining tests for the full sample, and for column 6 and 7 for the low educated, reduced form results are found to be positive but insignificant.

opportunities.

The low educated consistently shows larger positive effects of bank representation than the more highly educated do. This indicates that with the lack of formal education, capital access is greatly facilitated by co-ethnic bank contacts. The size of the OLS estimate means that exposure to at least one local co-ethnic bank employee increases self-employment probability by around 20 %. This could be due to the improved knowledge, or relatively favorable behavior, at the local bank where immigrants apply for loans. But it could also be an effect of the increased knowledge of the local financial system. This study cannot distinguish between these different mechanisms.

A wide range of sensitivity analyses have been performed, on the full sample and on the low educated, to make sure that the baseline analysis captures the relation between self-employment rate and co-ethnic bankers. The results are robust to changes in the definition of the explanatory variable and the outcome variable, as well as to changes in the model specification. This gives strong indications that the results are not driven by the specification and that the relation I estimate is stable. The model is also estimated studying both employees of the bank and the insurance sector, as these sectors are similar in many dimensions but differ in terms of supplying credit. The baseline results are robust to this, as they show much less of a relation between insurance company employees and self-employment than between bank employees and self-employment. Overall, this comparison suggests that the effect of co-ethnic bank employees differs from the generic effect on labor market outcomes of co-ethnics in well integrated sectors.

The results reported in this essay measure the effect of co-ethnic bankers on the self-employment probability. Without a measure of the actual capital accessed by the entrepreneurs, these measures serve as a proxy for the impact of shared background on capital access. Interesting avenues for future research would be to directly analyze the relationship between co-ethnic bankers and sources of finances. Due to data limitations, this has not been possible within the scope of this essay.

This essay has shown that the origin of bankers matters for the probability of immigrants from non-western countries to start up and run small businesses. It does not address the question of whether the facilitated capital access for immigrants is improving efficiency in the credit market. But future studies on survival rates, and characteristics of firms started, could add to the understanding of the ethnic component in the credit market, highlighted by the analyses presented above.

## References

- Aaronson, D., Bostic, R., Huck, P., and Townsend, R. (2004) "Supplier relationships and small business use of trade credit 55:46–67.
- Abbasian, S. and Yazdanfar, D. (2012) The Impact of Owner and Firm Characteristics on External Capital Acquisition at Start-up: Empirical Evidences from Swedish Data 5:19–30.
- Agency for Economic and Regional Growth (2007) Finansieringssituationen vid företagande for utrikes födda kvinnor och män.
- Agency for Economic and Regional Growth (2011) Företagare med utändsk bakgrund.
- Agency for Economic and Regional Growth (2012) Företagande i Sveriges regionerföretagens villkor och verklighet 2011.
- Andersson, R. (1998) Socio-Spatial Dynamics: Ethnic Divisions of Mobility and Housing in post-Palme Sweden 35:397–428.
- Andersson, R., Musterd, S., and Robinson, V. (2003) Spreading the 'burden'? A review of policies to disperse asylum seekers and refugees. The Policy Press, Bristol, UK.
- Andersson Joona, P. (2010) Exits from Self-Employment: Is There a Native-Immigrant Difference in Sweden? 44:539–559.
- Andersson Joona, P. and Wadensjö, E. (2008) A Note on Immigrant Representation in Temporary Agency Work and Self-employment in Sweden 22:495–507.
- Åslund, O., Edin, P.-A., and Fredriksson, P. (2003) Ethnic Enclaves and the Economic Success of Immigrants: Evidence from a Natural Experiment 118:329–357.
- Åslund, O., Edin, P.-A., Fredriksson, P., and Grönqvist, H. (2011) Peers, neighborhoods, and immigrant student achievement: Evidence from a placement policy 3:67–95.
- Åslund, O. and Fredriksson, P. (2009) Peer Effects in Welfare Dependence: Quasi-Experimental Evidence 20:798–825.
- Aslund, O., Hensvik, L., and Skans, O. N. (forthcoming) Seeking similarity: How immigrants and natives manage at the labor market .

- Åslund, O., Östh, J., and Zenou, Y. (2010) How important is access to jobs? Old question—improved answer 10:389–422.
- Bates, T. (1989) The Changing Nature of Minority Business: A Comparative Analysis of Asian, Non-Minority and Black-Owned Businesses 18:25–42.
- Blanchflower, D., Levine, B., and Zimmerman, D. (2003) Discrimination in the Small Business Credit Market 5:930–943.
- Blanchflower, D. and Oswald, A. (1998) What Makes an Entrepreneur? 16:26–60.
- Blume, K., Ejrnaes, M., Skyt Nielsen, H., and Wurtz, A. (2009) Labor market transitions of immigrants with emphasis on marginalization and self-employment 22:881–908.
- Borevi, K. and Myrberg, G. (2010) Välfardsstaten och de nyanlända: En flyktingplaceringspolitisk probleminventering.
- Bradbury, M. and Kellough, J. E. (2011) Representative Bureaucracy: Assessing the Evidence on Active Representation 41:157–167.
- Cavaluzzo, K. and Cavaluzzo, L. (1998) Market Structure and Discrimination: The Case of Small Businessess 30:771–792.
- Cavaluzzo, K. and Wolken, J. (2005) Small Business Loan Turndowns, Personal Wealth, and Discrimination 78:2153–2178.
- Committee on Foreign Born Entrepreneurship (1999) Invandrare som företagare- för lika möjligheter och ökad tillväxt- Betänkande av Utredningen om företagande för personer med utländsk bakgrund. SOU 1999:49. Stockholm fritzes.
- Evans, D. and Jovanovich, B. (1989) An Estimated Model of Entrepreneurial Choice under Liquidity Constraint 97:808–827.
- Evans, D. and Leighton, L. (1989) Some Empirical Aspects of Entrepreneurship 79:519–535.
- Fisman, R., Paraviini, D., and Vig, V. (2012) Cultural proximity and loan outcomes.
- Fraser, S. S. (2009) Is there Ethnic Discrimination in the UK Market for Small Business Credit? .
- Lindh, T. and Ohlsson, H. (1996) Self-employment and Windfall gains: Evidence from the Swedish Lottery\* 106:1515–1526.

Lindh, T. and Ohlsson, H. (1998) Self-employment and Wealth Inequality 44:25–41.

Phelps, E. S. (1972) The Statistical Theory of Racism and Sexism 62:659–661.

- Raijman, R. and Tienda, M. (2010) Ethnic foundations of economic transactions: Mexican and Korean immigrant entrepreneurs in Chicago 26:783–801.
- The Committee on Immigration Policy (1996) Sverige, framtiden och mångfalden. SOU 1996:55. Stockholm fritzes.
- The Immigration Board (1997) Individuell mångfald: Invandrarverkets utvärdering och analys av det samordnade flyktingmottagandet 1991-1996. Norrköping, Statens invandrarverk.

## 10 Appendix

#### 10.1 Variable description

- Bank Bank contains all firms defined as banks, but not other financial service companies. Industry coding is available from 1989 and onwards, so for the years 1987 and 1988 firms are given the industry coding it has 1989. This yields a possibility that firms that closed down before 1989 are not captured. Industry coding is also available at both firm and workplace level. Here I use workplace level industry coding in the baseline analysis but employ firm level industry coding in a sensitivity analysis.
- **Insurance sector** This includes insurance companies mainly focusing on injury insurances. The industry coding that it builds on is the same as the bank sector above.
- Self-employed Before 1993 only sole traders were counted as self-employed, and after 1993 also limited liability companies are included in this definition. For the baseline analysis I use a variable which uses the first definition the years before 1993 and after that the second definition. I furthermore only include self-employed with a positive annual income, and individuals for which self-employment income is the main income.
- Countries/regions of origin Some countries have a code of its own, as for example for Iran, while the whole of Latin America is categorized into 2 different codes. I use the 17 different countries/regions listed below.

<b>T</b> 11	0	a	c	• •
Table	8:	Countries	s of	origin
				- ()

-Bosnia and Herzegovina -Former Yugoslavia, Croatia, Macedonia, Slovenia (Balkan States) -Poland, Eastern Germany -Estonia, Lithuania, Latvia (Baltic states) -Romania, Former USSR, Bulgaria, Albania, Armenia, Azerbaijan, Georgia, Kazakhstan, Kirghizia, Moldova, Russia, Tadzhikistan's, Turkmenistan, Ukraine, Uzbekistan, Belarus (Eastern Europe) -Former Czechoslovakia, Hungary -Mexico, El Salvador, Other countries in the region (Central America) -Chile -Peru, Brazil, Columbia, Argentina, Uruguay, Other countries in the region (Latin America) -Ethiopia, Eritrea, Somalia, Sudan, Djibouti (Eastern Africa) -Lebanon, Syria, Morocco, Tunisia, Egypt, Algeria, Israel, Palestine, Jordan, Other countries in the region (Middle East) -Gambia, Uganda, Zaire, Ghana, Other countries in the region (Africa South of Sahara) -Iran -Iraq -Turkey -Fillipines, Indonesia, Malaysia, Singapore, Thailand, Vietnam, Other countries in the region (Southeast Asia) -Afghanistan, Bangladesh, Bhutan, India, Mongolia, Pakistan, Other

countries in the region (East Asia)

	All	Own-group banker=1	Own-group banker=0	
Bosnia & Hercegovina	0.0007	0.000	0.001	
Balkan States	0.0409	0.055	0.029	
Poland	0.0338	0.042	0.027	
Baltic states	0.0031	0.005	0.001	
Eastern Europe	0.0681	0.068	0.069	
Former Czechoslovakia	0.0224	0.025	0.020	
Central America	0.0139	0.005	0.022	
Chile	0.1036	0.111	0.097	
Latin America	0.0237	0.032	0.017	
Eastern Africa	0.1011	0.128	0.078	
Middle East	0.1496	0.102	0.191	
Africa South of Sahara	0.0207	0.021	0.020	
Iran	0.2480	0.272	0.228	
Iraq	0.0644	0.026	0.097	
Turkey	0.0343	0.038	0.031	
Southeast Asia	0.0457	0.032	0.057	
East Asia	0.0259	0.037	0.016	
Ν	53,518	24,683	28,835	

Table 9: Country of origin distribution 1996

#### 10.1.1 List of municipal group level covariates

Variables defined by those individuals from the own country who share the individual's municipality of residence

- Employment rate
- Share self-employed
- Share that group constitutes of local population
- Mean earnings
- Share with earnings above the median (defined by year)
- Share with university education
- Mean years since migration
- Share working in high income workplaces (defined as a workplace with median income above the yearly mean income)

Variables defined by mean values of workplaces where the local own-group members work (WP):

- Mean earnings at co-ethnics' workplaces
- Share with university education at co-ethnics' workplaces

Variables defined by co-ethnics' municipality of work (W):

- Mean earnings
- Share with earnings above the median (defined by year)
- Share with university education
- Share working in high income workplaces (defined as a workplace with median income above the yearly mean income)
- Mean earnings at co-ethnics' workplaces
- Share with university education at co-ethnics' workplaces

	All	Co-ethnic banker=1	Co-ethnic banker=0	
Share above median income (W)	0.2898	0.2982	0.2826	
Mean earnings $(1,000 \text{ SEK})(W)$	41.4301	42.6855	40.3554	
Share with university educ(W)	0.2858	0.2998	0.2737	
Share working on high income workplaces	0.3364	0.3378	0.3351	
Share working on high income workplaces(W)	0.3382	0.3390	0.3376	
Mean income $(1,000 \text{ SEK})$ (WP)	51.1261	52.8198	49.6762	
Mean income $(1,000 \text{ SEK})(W)$ (WP)	51.3051	52.9991	49.8550	
Share with university educ(WP)	0.2665	0.2973	0.2401	
Share with university educ(W) (WP)	0.2675	0.2991	0.2404	
N	53,518	24,683	28,835	

Table 10: Mean values of remaining municipality group level covariates, 1996

Note: (W) Variable defined by individuals who work in the municipality where one lives. (WP) Variable defined by the colleagues of an individuals' local co-ethnics.

## 10.2 Description; bank, insurance and self-employment

Education level	Full labor market	Bank sector	Insurance sector
Less than 9 years schooling	15.50	2.77	3.52
9 years schooling	17.85	7.28	7.42
High school, at most 2 years	22.91	25.87	20.48
High school, 3 years	19.62	24.28	26.85
University, at most 2 years	11.31	18.55	19.28
University, at least 3 years	11.72	20.90	21.93
PhD	1.08	0.34	0.52
Total	100.00	100.00	100.00

Table 11: Educational distribution among non-western employees in the bank and insurance sector



Figure 1: Income distribution- non western employees

Note: Income in 1,000 SEK. Deflated to 1980 years values.



Figure 2: Number of non-western employees in bank sector- over time

Note: Industry codes before 1989 are imputed from 1989 years values.



Figure 3: Number of self-employed non-western immigrants

Note: Before 1993 self-employment only includes sole traders, and from 1993 also limited liability companies are included



Figure 4: Share self-employed over years since migration

#### 10.3 Further regression results

Table 12: Estimated effects of exposure to co-ethnic banker on self-employment probability: instrumental variable and reduced form, further subsamples

	(1)	(2)	
	More than high school	Female	
Panel A. Reduced form			
Co-ethnic banker, year $t_0$	0.000617	-0.00181	
	(0.00352)	(0.00154)	
Observations	171,651	333,673	
R-squared	0.044	0.022	
Panel B. IV			
Co-ethnic banker, year t	0.00249	-0.00699	
	(0.0142)	(0.00599)	
Observations	171,651	333,673	
R-squared	0.044	0.021	
Panel C. First stage			
Co-ethnic banker, year $t_0$	0.248***	$0.259^{***}$	
	(0.0202)	(0.0203)	
Observations	171,651	$333,\!673$	
R-squared	0.318	0.344	

Note: Panel A reports estimated  $\beta$  parameters (standard errors) from equation (2), while Panel B reports estimated  $\beta$  parameters (standard errors) from a version of equation (1) where the endogenous regressor Co-ethnic banker<sub>it</sub> is instrumented by equation (3). Panel C reports estimated  $\beta$  parameters (standard errors) from the first stage regression, equation (3). The outcome variable is a dummy for being self-employed in panel A and B, while it is a dummy for the exposure to a co-ethnic banker in year t in panel C. Standard errors are clustered at the level of municipality of arrival and country of origin pairs.\*\*\* p<0.01, \*\* p<0.05, \* p<0.1. Fixed effects for year-, year of immigration-, municipality of arrival- and country of origin are included as well as controls for sex, age and education level. A full set of municipality of arrival group level covariates are also included, see list in the appendix for overview of these variables.

	All		Less than high school	
	(1)	(2)	(3)	(4)
	Entry	Exit	Entry	Exit
Co-ethnic banker	$0.00134^{*}$	-0.0159	$0.00136^{*}$	-0.0306
	(0.000713)	(0.0106)	(0.000814)	(0.0207)
Co-ethnic insur- ance employee	0.00121*	-0.0175	0.00146*	-0.0225
	(0.000664)	(0.0114)	(0.000887)	(0.0198)
Observations	652 370	22 246	263 923	8 607
R-squared	0.013	0.039	0.018	0.054

Table 13: Effects of co-ethnic bankers and insurance company employees on probability of start-up and self-employment exit: OLS

Note: The table reports  $\beta$  parameters (standard errors) from a version of equation (1), where both Co-ethnic bankers and own-group insurance company employees are used as explanatory variables. The outcome variable used in column 1 and 3 (2 and 4) is a dummy for becoming selfemployed (exiting self-employment). Standard errors are clustered at the level of municipality and country of origin pairs.\*\*\* p<0.01, \*\* p<0.05, \* p<0.1 Fixed effects for year-, year of immigration-, municipality- and country of origin are included, as well as controls for sex, age and education level. Municipal group level covariates are contemporary, see list in appendix for overview over these variables.

	(1)	(2)
	All	Less than high school
	0.00500***	0.00040***
Co-ethnic banker	0.00580	(0.00188)
	(0.00148)	(0.00188)
Employment rate	0.0347***	0.0397***
	(0.00610)	(0.00809)
Mean earnings	1.44e-05	3.34e-05
	(1.60e-05)	(2.22e-05)
Share with above median income	0.00849	-0.00837
	(0.0118)	(0.0166)
Share of local population	0.00962	-0.169
	(0.127)	(0.131)
Mean years in country	$-0.000741^{**}$	-0.000737
	(0.000330)	(0.000477)
Share with above median income (W)	0.00172	0.00440
	(0.00869)	(0.0126)
Mean earnings (W)	2.76e-06	9.77e-06
	(1.22e-05)	(1.76e-05)
Share self-employed	0.353***	0.337***
	(0.0268)	(0.0336)
Municipality size	-8.48e-08	-1.27e-07
	(6.52e-08)	(7.98e-08)
Share with university education (W)	0.00231	0.00205
	(0.00626)	(0.00880)
Share with university education	0.00618	0.00697
-	(0.0101)	(0.0139)
Share with university educ (W) (WP)	0.0416***	0.0496**
• • • • • • • •	(0.0153)	(0.0227)
Mean income (W) (WP)	-9.20e-05***	-9.36e-05***
	(1.84e-05)	(2.55e-05)
Share working in high income workplaces (W)	0.0109	0.00891
5 5 · · · · · · · · · · · · · · · · · ·	(0.00726)	(0.00991)
Mean income (WP)	-9.50e-05***	-0.000102***
	(1.89e-05)	(2.67e-05)
Share with university educ(WP)	0.00102	-0.0330
	(0.0156)	(0.0211)
Share working in high income workplaces	0.0195**	0.0123
Share working in high meone workplaces	(0.00830)	(0.0106)
Observations	800,899	325,430
R-squared	0.053	0.067

Table 14: Detailed regression results of column 1 and 2 in table 2: OLS

Note: The table reports estimated  $\beta$  parameters (standard errors) from equation (1). The outcome variable is a dummy for being self-employed. Robust standard errors are clustered at the level of each municipality country of origin pair. \*\*\* p<0.01, \*\* p<0.05, \* p<0.1. Fixed effects for year-, year of immigration-, municipality- and country of origin are included as well as controls for sex, age and education level. (W) Variable defined by individuals who work in the municipality where one lives. (WP) Variable defined by the colleagues of an individuals' own-group members.

#### Publication series published by IFAU – latest issues

#### **Rapporter/Reports**

- 2014:1 Assadi Anahita "En profilfråga: Hur använder arbetsförmedlare bedömningsstödet?"
- 2014:2 Eliason Marcus "Uppsägningar och alkoholrelaterad sjuklighet och dödlighet"
- **2014:3** Adman Per "Försummas gymnasieskolans demokratiuppdrag? En kvalitativ textanalys av 2009 års svenska gymnasiereform"
- **2014:4** Stenberg Anders and Olle Westerlund "Utbildning vid arbetslöshet: en jämförande studie av yrkesinriktad och teoretisk utbildning på lång sikt"
- 2014:5 van den Berg Gerard J., Lene Back Kjærsgaard and Michael Rosholm "Betydelsen av möten mellan arbetslösa och förmedlare"
- 2014:6 Mörk Eva, Anna Sjögren and Helena Svaleryd "Blir barn sjuka när föräldrarna blir arbetslösa?"
- **2014:7** Johansson Per, Arizo Karimi and J. Peter Nilsson "Könsskillnader i hur sjukfrånvaro påverkas av omgivningen"
- **2014:8** Forslund Anders, Lena Hensvik, Oskar Nordström Skans, Alexander Westerberg and Tove Eliasson "Avtalslöner, löner och sysselsättning"
- 2014:9 Engdahl Mattias "Medborgarskap, arbetsmarknaden och familjebildning"
- 2014:10 Hallberg Daniel, Per Johansson and Malin Josephson "Hälsoeffekter av tidigarelagd pensionering"
- 2014:11 Karbownik Krzysztof and Sara Martinson "Svenska högstadie- och gymnasielärares rörlighet på arbetsmarknaden"
- **2014:12** Hägglund Pathric, Per Johansson and Lisa Laun "Insatserna inom rehabiliteringsgarantin och deras effekter på hälsa och sjukfrånvaro"
- 2014:13 Regnér Johan "Effekter av yrkesinriktad arbetsmarknadsutbildning för deltagare med funktionsnedsättning, 1999–2006"
- **2014:14** Assadi Anahita and Martin Lundin "Enhetlighet och träffsäkerhet i arbetsmarknadspolitiken: Hur använder arbetsförmedlare statistisk profilering i mötet med den arbetssökande?"
- **2014:15** Edmark Karin, Markus Frölich and Verena Wondratschek "Hur har 1990-talets skolvalsreformer påverkat elever med olika familjebakgrund?"
- 2014:16 Karimi Arizo "Sen familjebildning, täta födelseintervall och kvinnors inkomster"
- 2014:17 Eliasson Tove "Bankanställdas ursprungsland och egenföretagande bland utrikesfödda"

#### Working papers

- **2014:1** Vikström Johan "IPW estimation and related estimators for evaluation of active labor market policies in a dynamic setting"
- **2014:2** Adman Per "Who cares about the democratic mandate of education? A text analysis of the Swedish secondary education reform of 2009"
- **2014:3** Stenberg Anders and Olle Westerlund "The long-term earnings consequences of general vs. specific training of the unemployed"
- **2014:4** Boye Katarina "Can you stay at home today? The relationship between economic dependence, parents' occupation and care leave for sick children"
- **2014:5** Bergemann Annette and Gerard J. van den Berg "From giving birth to paid labor: the effects of adult education for prime-aged mothers"
- **2014:6** van den Berg Gerard J., Lene Kjærsgaard and Michael Rosholm "To meet or not to meet, that is the question short-run effects of high-frequency meetings with case workers"
- **2014:7** Avdic Daniel, Petter Lundborg and Johan Vikström "Learning-by-doing in a highly skilled profession when stakes are high: evidence from advanced cancer surgery"

- 2014:8 Mörk Eva, Anna Sjögren and Helena Svaleryd "Parental unemployment and child health"
- **2014:9** Johansson Per, Arizo Karimi and J. Peter Nilsson "Gender differences in shirking: monitoring or social preferences? Evidence from a field experiment"
- **2014:10** Eliasson Tove and Oskar Nordström Skans "Negotiated wage increases and the labor market outcomes of low-wage workers: evidence from the Swedish public sector"
- 2014:11 Engdahl Mattias "Naturalizations and the economic and social integration of immigrants"
- **2014:12** Hallberg Daniel, Per Johansson and Malin Josephson "Early retirement and post-retirement health"
- 2014:13 Karbownik Krzysztof "The determinants of teacher mobility in Sweden"
- 2014:14 Karbownik Krzysztof "Job mobility among high-skilled and low-skilled teachers"
- **2014:15** Karbownik Krzysztof "Do changes in student quality affect teacher mobility? Evidence from an admission reform"
- **2014:16** Edmark Karin, Markus Frölich and Verena Wondratschek "Sweden's school choice reform and equality of opportunity"
- **2014:17** Karimi Arizo "Effects of the timing of births on women's earnings evidence from a natural experiment"
- 2014:18 Karimi Arizo "The spacing of births and women's subsequent earnings evidence from a natural experiment"
- 2014:19 Eliasson Tove "Immigrant entrepreneurship and the origin of bankers"

#### **Dissertation series**

- 2013:1 Vikman Ulrika "Benefits or work? Social programs and labor supply"
- 2013:2 Hanspers Kajsa "Essays on welfare dependency and the privatization of welfare services"
- 2013:3 Persson Anna "Activation programs, benefit take-up, and labor market attachment"
- 2013:4 Engdahl Mattias "International mobility and the labor market"